An-Najah National University Faculty of Graduate Studies

Liquidity Risk Management Practices: A comparative study between Islamic and Conventional Banks in Palestine

By Raghad Naser Ali Al-Ashqar

Supervisor Dr. Ra'fat Al-Jallad

This Thesis is Submitted in Partial Fulfillment of the Requirements for the Degree of Master of Finance, Faculty of Graduate Studies, An-Najah National University. Nablus, Palestine.

2020

Liquidity Risk Management Practices: A comparative study between Islamic and Conventional Banks in Palestine

By Raghad Naser Ali Al-Ashqar

This Thesis was defended successfully on 2/9/2020, and approved by:

Defense Committee Members

Dr. Ra'fat Al-Jallad \ Supervisor
Dr. Suliman Al-Abbady \ External Examiner Suleiman. Albade

3. Dr. Islam Abd Al-Jawad \ Internal Examiner

Signature

Acknowledgment

In the name of God, the Most Gracious, the Most Merciful. We praise you, we thank you for all blessings that you have given to us from a thoughtful mind and a speaking tongue to express all we seek and found in our minds, we pray and greet our noble messengers, Muhammad, (peace be upon him) who taught the nation and carried the message, I extend my sincere gratitude to everyone who has the most sincere intention in education; you have enlightened our minds, and educated generations, I thank my supervisor and my dear professor, Dr. Ra'fat Al-Jallad, who was accompanying me at every step, who has always dedicated himself to provide support to me, who gave me the best guidance, and deserves nothing from us except every supplication and good, I thank professors who never refused my help or support. I offer this work to my family that gave me love and affection, who I was always in their priorities, for those who made me what I am today and fill me with proud that I am their daughter, I ask God to protect them, I thank everyone I didn't mention specifically in my words, thank you for helping me, I hope you like it as a result of my tiredness, I also dedicate this work to my lifetime friend and companion, Derby Nour, who supported me in every step, shared my joy, my pain, my happiness, and my sorrow, she was the healing balm for me and present all the time and offered me encouragement and help. May God protect and perpetuate her...Finally, to my son Ilyas, who is the greatest gift of the Most Merciful God to me, to him, I direct my tiredness, my effort, and my work, I will always dedicate myself to make him proud of me more and more, who will be an incentive for me to work hard and strive, I ask God to prolong his life, to protect him from all evil and progress and supremacy will always be his ally.

الاقرار

انا الموقع ادناه مقدم الرساله التي تحمل العنوان:

Liquidity Risk Management Practices: A comparative study between Islamic and Conventional Banks in Palestine

ممارسات ادارة مخاطر السيوله: دراسه مقارنه بين

البنوك التقليديه والاسلاميه في فلسطين

أقر بأن ما اشتملت عليه هذه الرساله انما هي نتاج جهدي الخاص،باستثناء ما تمت الاشاره اليه حيثما ورد، وأن هذه الرسالهككل، أو أي جزء منها لم يقدم لنيل أي درجة علميه او لقب علمي او بحثي لدى اية مؤسسة تعليميه اخرى

Declaration

The work provided in this thesis, unless otherwise referenced, is the researcher's own work and has not been submitted elsewhere for any other degree or qualification

Student's name:

Signature:

اسم الطالبة: رغ رئلم مرع) المن شقر التوقيع: وغر المأ مشتر التاريخ: 2/9/20

Date:

Table of contents

No.	Contents	Page
	Acknowledgment	Iii
	Declaration	Iv
	Table of contents	V
	List of Tables	Vii
	List of Figures	viii
	List of Annexes	Ix
	Abstract	Х
	Chapter One: Introduction	1
1.1	Overview	2
1.2	Problem of the study and Research Questions	6
1.3	Significance of Research	6
1.4	Research objectives	7
1.5	Research structure	8
	Chapter Two: Palestinian Banking System	9
2.1	Introduction	10
2.2	Bank's Characteristics	10
2.2.1	Introduction	10
2.3	Bank's indicators	11
2.3.1	Profitability in Palestinian banks	11
2.3.2	Deposits in Palestinian banks	13
2.3.3	Geographic distribution of deposits and credit facilities	14
2.3.4	Public and private loans	16
2.3.5	Capital adequacy ratios in Palestinian banks	16
2.3.6	Liquidity ratios in Palestinian banks	17
	Chapter Three: Literature Review	20
3.1	Conceptual Framework	21
3.1.1	The Concept of Liquidity Risk	21
3.1.2	Islamic Finance Principles	24
3.1.3	Modes of Finance in Islamic Banks	26
3.1.4	Risks Facing Conventional and Islamic banks	28
3.1.5	Liquidity Risk Management Process	34
3.2	Hypotheses development	38
3.3	Functional framework	39
	Chapter Four: Research Methodology	41
4.1	Quantitative Research Methodology	42
4.1.1	Data and Sample Descriptions	42
4.1.2	Model Specification & Variables Definition	43

No.	Contents	Page		
4.1.3	Variables Measurement	44		
4.2	Qualitative Research Methodology	47		
4.2.1	Data and Sample Description	47		
4.2.2	Questionnaire Design	48		
4.3	Propositions	48		
	Chapter Five: Results and Discussion	50		
5.1	Preface	51		
5.2	Quantitative approach	51		
5.2.1	Summary Statistics	51		
5.2.2	Correlation analysis			
5.2.3	Model Estimation			
5.2.4	Discussion of Estimation Results for Model One	55		
5.2.5	Discussions of Estimation Results for Model Two	58		
5.4	Qualitative Results and Discussions	60		
	Chapter Six: Recommendations and Future Research Directions	64		
	References	67		
	Annexes	81		
	الملخص	Ļ		

No.	Table	Page			
Table (2.1)	Banks operating in Palestine 2018	11			
Table (2.2)	Banking Indicators	12			
Table (2.3)	deposits in conventional banks	13			
Table (2.4)	deposits in Islamic banks	14			
Table (2.5)	Geographic distribution of credit facilities	15			
Table (2.6)	liquidity ratios in conventional banks	18			
Table (2.7)	liquidity ratios in Islamic banks	18			
Table (2.8)	liquidity ratios in combined Islamic and conventional banks	18			
Table (4.1)	Measurement of Variables				
Table (4.2)	Sample description				
Summary statistics for Islamic and conventional banks					
Table(5.2)	Full sample correlation matrix				
Table(5.3)	Estimation results for liquidity risk	54			

List of Tables

vii

viii
List of Figures

No.	No. Table				
Figure (2.1)	Geographic distribution of deposits	15			
Figure (2.2)	Public and Private Loans	16			
Figure (2.3)	Capital Adequacy Ratio in Palestinian banks 2018	17			
Figure (3.1)	Islamic and conventional banking risks overview	29			
Figure (3.2)	igure (3.2) Liquidity risk management process in banks				
Figure (3.3)	Conceptual frame work	40			

List of Annexes

No.	Annex	Page
Annex (1)	Questionnaire to Islamic Banks	82
Annex (2)	Questionnaire to conventional Banks	83

Liquidity Risk Management Practices: A comparative study between Islamic and Conventional Banks in Palestine By Raghad Naser Ali Al-Ashqar Supervisor Dr. Ra'fat Al-Jallad

Abstract

Banks face numerous kinds of risks including liquidity risk. The issue of liquidity is the most important issue in the banking sector (conventional and Islamic), where the bank may lose customers as a result of the lack of sufficient liquidity, or the inability to meet their withdrawals in a timely manner. Moreover, banks are accountable toward the supervisory authorities, which affects the reputation of the financial institution in case of liquidity shortage. On the other hand, the bank may maintain more liquidity than it needs, resulting in a situation of inefficient use of available resources. Therefore the existence of proper liquidity management increases the confidence of the supervisory authorities and depositors. With the financial institutions, conventional banks can utilize numerous devices to manage this sort of hazard. Islamic banks, however, are constrained in utilizing portion of these instruments. In this manner, it is critical to comprehend the idea of liquidity risk in Islamic and conventional banks and what variables can influence it. A mixed research approach methodology was used in this thesis. The qualitative approach was conducted with a sample of 3 Islamic banks and 3 conventional banks. Open questions were used to collect data from Islamic and conventional bankers. A quantitative research methodology was used to model the liquidity risk in Islamic and conventional banks of Palestine. Capital adequacy ratio (CAR), return on equity (ROE), deposits to assets (DA), bank size (BS), and bank nationality has been used to test their impact on liquidity risk for both Islamic and conventional banks over the period 2009-2018. The finding of this thesis has revealed that there is a positive relation between deposits to assets and liquidity risk, a negative and significant relation between capital adequacy ratio and liquidity risk for the banks in Palestine. While the effect of interaction variable "Islamic bank" on the effect of DA, ROE and CAR on liquidity were found to be significant. Findings of this paper will help banks' management to decrease liquidity risk and keep their banks at a better liquidity position. Banks managers should be increasingly cautious when financing their customers as expanding financing can deteriorate liquidity. Some recommendations of this thesis involve that banks should have integrated liquidity management framework and increase their investment portfolio because the existence of the liquidity issue with banks means not to fully employ the sources of funds. Moreover, banks should have a specialized liquidity division. Furthermore, Palestine Monetary Authority, though stand prepared to help and bailout any bank confronting liquidity issues, should enforce Israeli side to remit the Shekel currency; hence, it must review the current relationships with Israeli banks and find a new sustainable mechanism that serves the interests of the Palestinian banking.

Chapter One Introduction

Chapter One Introduction

1.1 Overview

The financial markets in general and the banking industry in particular, have been generally perceived as indicators of the state of the development of any country by government officials and academics. The rise of modern Islamic banking started with the experience of a local savings bank that appeared in Egypt in 1963 by Dr. Ahmed Al-Najjar. This attempt was based on collecting personal savings from depositors and investing them (Soylu, 2019).

The formal Islamic banking business started in 1975, when Dubai Islamic Bank (DIB) was established in the United Arab of Emirates. At the same year, the Islamic Development Bank (IDB) was established in Jeddah (Soylu, 2019). After 1975, the number of Islamic banks increased dramatically. By the year of 2019, number of Islamic banks was 520, number of Takaful operation was330, number of Sukuk issuances was 12000, number of Islamic funds was 1700 and number of Islamic finance was 960. With a total asset of 2.524 trillion USD (IFSB, 2019).

The banking sector performs a very crucial financial intermediation. Through facilitating the flow of capital to various sectors of the economy Arif and Anees (2012). The Palestinian banking sector experienced a rapid development, due to technological changes, competition, liberalization, and restructuring introduced by the Palestinian Monetary Authority (PMA), to improve the regulatory framework and ensure the well-being of the banking sector.

Every bank faces operational risk that is related to the internal operations of the bank. Business risk, on the other hand, is associated with the industry risk, which relates to the external business environment of the banks. Lastly, financial risk is related to the probability of default of a bank. The financial risk of banks can be further divided into three categories: market operational risk, credit risk and liquidity risk (Keramati & Shaeri, 2014).

Rosely (2008) stated that liquidity and profitability both are most bulbous issues in the realm of corporate finance. The term of liquidity is defined by Ibe (2013) as the degree of convertibility to cash or in other words, the ease with which any assets can be converted to cash is called liquidity in financial literature. Durrah (2016) reported that liquidity of banks refers the banks' ability to satisfy its short terms financial obligations and requirements when they come due.

Liquidity risk is the potential loss to a bank arising from its inability either to meet its obligations, or to fund increase in assets as they fall due without incurring unacceptable costs (IFSB, 2012). The main reason that lead the bank inherently faces liquidity risk is the maturity mismatch between the bank's assets and liabilities due to transforming short term deposits into long term loans and investments. The fact that Basel standards regarding liquidity management were inconsistent with Islamic banks, led to introducing special standards by Islamic Financial Services Board. Because of the importance of a sound liquidity risk management framework for any well-functioning institution offering Islamic financial services. Islamic Financial Institutions have often found management of liquidity risk a very challenging part of their operations. These challenges can be observed at all tiers of the liquidity risk management framework of an IIFS – institutional, interbank and central bank levels.

Based on the existence of such standards and through reviewing previous literature, we are not aware of studies that dealt with the difference in liquidity risk management between Islamic and conventional banks, or the factors that impact the liquidity risk in the banking sector in Palestine.

With regard to the issue of managing liquidity risk, previous literature from Palestine dealt with the role of accounting and financial information in managing liquidity risk for commercial banks in Gaza. The study found that commercial banks use accounting and financial information in forecasting liquidity risk. Commercial banks use special reports and financial ratios to identify the liquidity available. Senior management and regulators use follow-up and monitoring reports on a regular basis to help in detecting deviations concerning liquidity position and correct it (Ahlasa & Ramadan, 2013).

Another study dealt with the impact of risk management on the degree of safety in the Palestinian banking system. The study showed that there was a direct proportion between the banking security level, on the one hand, and all risks related to liquidity level, lack of capital, fluctuation of interest rates and return on assets, on the other hand (Shaheen, Sabah & Misbah, 2011). Some other studies dealt with the effect of the variables used in this thesis on profitability and on credit risk (Abbadi & Karash, 2013; Abugamea, 2018; Rosman, 2009, Al-Darraji, 2015).

The studies conducted in Palestine related liquidity risk in Islamic and conventional banks are slightly limited. Abdelkarim and Burbar (2007) focus on concepts in risk management and financial risk management in Palestinian conventional and Islamic banks. The study showed that banks in Palestine have poor risk management system for monitoring, measuring, and mitigating risks. There is no department responsible to identify, monitor, and manage financial risks. Most banks do not have formally written approved and circulated work policies. Banks in Palestine lack adequate instruments for managing the risks they faced.

Bayyoud and Sayyad (2015) focus on the role of risk management on banks in Palestine. The findings showed that overall internal control and risk management system in Palestine have positive effect on bank's performance.

Therefore, the literature in the Palestinian context lacks empirical evidence on the differences in liquidity risk management practices between

5

Palestinian Islamic and conventional banks. The objective of this thesis is to fill the gap in the literature through investigating the differences between liquidity risk management in Islamic and conventional banks and provide guidelines to manage such risk across these banks.

1.2 Problem of the Study and Research Questions

The differences between the business models of Islamic and their conventional counterparts play an important role in managing liquidity risk. However, do these differences in Islamic and conventional bank's business model translated into differences in risk management practices related to liquidity risk? To answer this main research question, we need to answer the following sub questions:

- 1. How banks measure and manage liquidity risk in Palestine?
- 2. What kind of liquidity risk exists in Palestinian Islamic and conventional banks?
- 3. What are the factors contributed to liquidity risk management in Palestinian banks?
- 4. How the type of bank, whether Islamic or conventional, affect the relationship between factors determine liquidity and liquidity risk management practices?

1.3 Research objectives

This thesis aims to achieve the following objectives:

- Defining the differences in liquidity risk measurement and management between Islamic and conventional banks.
- 2. Exploring the types of liquidity risk in Palestinian Islamic and conventional banks.
- 3. Investigating the factors contributed to liquidity risk management in Palestinian banks?
- 4. Investigating the effect of the type of bank, whether Islamic or conventional, on the relationship between factors determine liquidity and liquidity risk management practices?

1.4 Significance of Research

The importance of this research is to focus on the Islamic finance industry in Palestine. In light of the growing importance of Islamic banks in the global financial system and Shariah-compliant financial innovations. The importance of sound risk management practices. As in most emerging markets, the Palestinian banking sector dominates the financial sector as financings are channeled through banks (Islamic and conventional) and not through the capital market. This fact exposes the entire economy to the systematic risk of bank failure.

Answering thesis questions will help banks' directors to decrease liquidity risk. Provide guidelines to manage such risks across these banks.

1.5 Research Structure

This thesis structured as follow: chapter 1 is the introduction, while chapter 2 covers the background about banking sector in Palestine. Chapter 3 presents the literature review of previous studies. Chapter 4 explains the methodology, data and variables. Furthermore, chapter 5 presents the analysis and results of this study. Finally, chapter 6 appertain conclusion.

Chapter Two Palestinian Banking System

Chapter Two Palestinian Banking System

2.1 Introduction

The financial sector in Palestine has emerged since signing Oslo Accord in 1993. After Paris Protocol in 1994, this allowed the Palestinian Monetary Authority (PMA) to administer monetary and financial affairs.

Palestinian Monetary Authority seeks to increase the efficiency of the banking services provided to customers. Through the branching policy that aims to reduce the population density of the number of branches and to become more in line with the internationally recognized rates (10,000 people per branch) (PMA, 2018).

In this context, the population index recorded for each branch showed some improvement in 2018, dropping from 12.8 in 2017 to 6.12 in 2018. Despite this continuous improvement in the ratio, there is still room for further branching (PMA, 2018).

2.2 Bank's Characteristics

2.2.1 Introduction

The year 2018 witnessed a change in the structure of the licensed banks. As a result of Al-Quds bank acquisition of Jordan Kuwait bank in Palestine, thus the number of licensed banks decreased from 15 to 14 banks, of which: 3 local conventional banks, 3 local Islamic banks and 8 foreign banks (7 Jordanian banks and 1 Egyptian bank). In 2018, numbers of the banking branches and offices have increased by 14 branches to reach 351 branches in compared to 337 branches in 2017 (PMA, 2019).

Table 2.1 lists of the banks operating in Palestine and their general characteristics.

Bank name	Bank type	Public/	Listing	# of	# of
	T 1 ·	Private	market	branches	employees
Safa Bank	Islamic	Private	Palestine	6	109
Palestine Islamic Bank	Islamic	Private	Palestine	43	664
Arab Islamic Bank	Islamic	Private	Palestine	22	531
Jordan Commercial Bank	Conventional	Private	Jordan	5	122
Egyptian Arab Land Bank	Conventional	Private	Egypt	7	NA
Jordan Ahli Bank	Conventional	Private	Jordan	9	218
Palestine Investment Bank	Conventional	Private	Palestine	18	243
Bank of Jordan	Conventional	Private	Jordan	36	NA
Housing Bank for Trade & Finance	Conventional	Private	Jordan	15	279
Cairo Amman Bank	o Amman Bank Conventional		Jordan	21	2049
Bank of Palestine	Conventional	Private	Palestine	72	1300
The National Bank	National Bank Conventional		Palestine	26	592
Al-Quds Bank	Al-Quds Bank Conventional		Palestine	40	747
Arab Bank	Conventional	Private	Jordan	31	897

Table (2.1): Banks operating in Palestine 2018

Source: Association of banks in Palestine. Available at https://abp.ps/

2.3 Bank's Indicators

2.3.1 Profitability in Palestinian banks

Most banking income still comes from traditional lending activities. It all ties back to the fundamental way banks make money. Banks use depositors' money to make loans. The amount of interest the banks collect on the loans is greater than the amount of interest they pay to customers with saving accounts. The difference is the banks profit.

Table 2.2 reports the profit, loans and assets for the banks operating in Palestine over the period 2013 to 2018.

Year	2013	2014	2015	2016	2017	2018	Compound annual growth rate (CAGR)
Profit (USD million)	143	147	132	149	168	176	
Growth		2.4%	-9.6%	12.4%	13%	4.6%	4.24%
Loans (USD million	4404	4816	5736	6785	7900	8189	
Growth		9.4%	19.1%	17.9%	18.8%	3.7%	13.2%
Assets (USD million)	10782	11416	12271	13765	15376	15475	
Growth		6%	7.5%	12.2%	12.3%	1.0%	7.49%

Table (2.2): Banking Indicators

Source https://www.abp.ps/

As the table above shows, profit of the banking sector for Palestinian listed banks amounted to 143 USD million in 2013, grew by a compound average annual rate of 4.24% to reach 176 USD million in 2018. The assets of the Palestinian banking sector have increased from 1078 USD million in 2013 to 1547 USD million in 2018. The increase in assets value across these years is due to increase in loans as loans represent 45% to the total banking assets system. Loans consider the main profitability source for banks in general. Loans granted by the Palestinian banking sector amounted to 4404 USD million in 2013 and increased during the period to reach 8189 USD million in 2018.

2.3.2 Deposits in Palestinian banks

The success of a bank depends on its ability to attract deposits, particularly time deposits. As it can be used for lending, financing or long-term investing. The deposits of Palestinian banks reached 12.193 million in 2018 USD (PMA, 2018). The main component of banks deposits is customer deposits (saving and time), which accounts for 93% of the total liabilities of the banking sector, other component of deposits are: PMA deposits, deposits from local banks and deposits from foreign banks outside Palestine (PMA, 2019).

The PMA attributes this increase in deposits to the branching policy PMA (2019). Opening Banks branches in villages and rural areas has enhanced customer ability to complete their financial transaction. The following Table 2.3 reports the consolidated deposits for conventional banks.

Year	2013	2014	2015	2016	2017	2018	Compound annual growth rate (CAGR)
Deposits (USD million)	7836	8258	8954	8169	12009	13076	
Growth		6%	7.5%	-8.7%	47%	8.8%	8.5%

Table (2.3): deposits in conventional banks

Source <u>https://www.abp.ps/</u>

The level of deposits for the conventional banks amounted to 7836 USD million in 2013 and grew by 8.5% to 13076 USD million in 2018 (ABP, 2014).

Year	2013	2014	2015	2016	2017	2018	Compound annual growth rate (CAGR)
Deposits (USD million)	840	994	1152	1398	1822	1993	
Growth		18.3%	15.8%	21.3%	30.3%	9.3%	15.4%

Table (2.4): deposits in Islamic banks

Source <u>https://www.abp.ps/</u>

The level of deposits for the Islamic banks amounted to 840 USD million in 2013 and have a positive growth by 15% to 1993 USD million in 2018 (see Table 2.4). Deposits of conventional banks were much higher than deposits of Islamic banks due to the greater number of conventional banks over Islamic banks.

2.3.3 Geographic distribution of deposits and credit facilities

Figure 2.1 presents the distribution of customer deposits by geographic area. Ramallah has the largest portion of deposits (35%) followed by Nablus (13%). The distribution of deposits is parallel to that of credit facilities. Table 2.5 shows the geographic distribution of credit facilities. The majority of loans granted in Ramallah (52%) followed by Nablus (12%). This high portion of credit facilities in Ramallah is due to the fact that most of government, private sector and NGO offices are located there. This allowed to private and public sectors employees living

in the city of Ramallah, which increase the demand for consumer goods, real estates...etc.



Figure (2.1): Geographic distribution of deposits

Source; Association of banks in Palestine. Available at https://www.abp.ps/

Governorate	Percentage				
Nablus	12%				
Jenin	3.2%				
Tulkarem	4%				
Qalqelia	1.20%				
Salfeet	.70%				
Tubas	.50%				
Ramallah	52%				
Jerusalem	3.20%				
Jericho	1.85%				
Hebron	6%				
Bethlehem	5.70%				
Gaza strip	10%				
Sources https://www.ahn.ng/					

Source; https://www.abp.ps/

2.3.4 Public and private loans

Figure 2.2 shows the trend of public and private loans provided by the banking sector. Although recent years have witnessed a remarkable decline in the size and importance of the credit portfolio granted to the public sector, it still constitutes an insignificant part of the credit portfolio (15.6% in 2018 comparing with 18.4% in 2017 (PMA, 2018).



Figure (2.2): Public and Private Loans Source: Association of banks in Palestine. Available at <u>https://www.abp.ps/</u>

2.3.5 Capital adequacy ratios in Palestinian banks

According to PMA Instruction number 7 for the year 2016, Banks operating in Palestine are required to keep a minimum capital adequacy (CAR) (PMA, 2016). This ratio is being adjusted according to bank size potential risks. This ratio is based on the bank size of operation, lending rate and the value of risk weighted assets. Al-Safa bank operates with high CAR of 58.40%, Egyptian Arab land bank also with 39%. Higher CAR reflects less risky weighted assets and indicates inefficient use of resources (Abu Zir, 2016). However, Arab bank operate with the least CAR 12.44% and Palestine Islamic operates with a low CAR of 12.81% because of the significant lending portfolio of the two banks.

Figure 2.3 represents the values of CAR for the banks operating in Palestine as of 2018.



Figure (2.3): Capital Adequacy Ratio in Palestinian banks 2018 Source: Association of banks in Palestine. Available at <u>https://www.abp.ps/</u>

2.3.6 Liquidity ratios in Palestinian banks

Table 2.6 and 2.7 provide a comparison in liquidity ratios between conventional and Islamic banks as follows:

Conventional banks						
Liquidity ratios	2013	2014	2015	2016	2017	2018
Customer deposits/Total assets	6.36%	6.44%	6.64%	6.55%	6.70%	6.78%
Net credit facilities/Total assets	3.91%	3.99%	.33%	.48%	4.52%	4.51%
Net direct credit facilities/Customer deposits	5.5%	5%	6.01%	6.03%	6.04%	6.7%

Table (2.6): liquidity ratios in conventional banks

Source: Association of banks in Palestine. Available at <u>www.ps/index.</u>

Table (2.7): liquidity ratios in Islamic banks

Islamic banks						
Liquidity ratios	2013	2014	2015	2016	2017	2018
Customer	.387%	.381%	.395%	11%	21%	22.93%
deposits/Total assets						
Net credit	.236%	.258%	.285%	.139%	.173%	.198%
facilities/Total assets						
Net direct credit						
facilities/Customer	30%	34%	36%	22%	38%	30.3%
deposits						

Source: Association of banks in Palestine. Available at www.ps/index.

Table (2.8): liquidity ratios in combined Islamic and conventional banks

Conventional banks and Islamic banks						
Liquidity ratios	2013	2014	2015	2016	2017	2018
Customer	76.8%	78%	78.5%	77.3%	78%	78.8%
deposits/Total assets						
Net direct credit	40.8%	42.2%	46.8%	49.4%	51.5%	52.9%
facilities/Total assets						
Net direct credit						
facilities/Customer	53.2%	54.1%	59.6%	63.9%	66.1%	67.2%
deposits						

Source: Association of banks in Palestine. Available at <u>www.ps/index.</u>

Liquidity ratios are explained as follows:

First of all, liquidity is associated with the deposits to assets ratio (DA): this ratio measures the magnitude of assets being funded by

customer's deposits. Higher ratio means higher level of liquidity risk, because these deposits are subjected to sudden withdrawals at any time (Mwangi, Muturi and Ombuki, 2015).

Secondly, liquidity is also associated with the percentage of net direct credit facilities from the total assets. Granted loans constitute the main source of bank profits. Higher percentage of this ratio indicates a higher capacity of the bank's ability to deal with liquidity problems or sudden withdrawals (Barnes, 1987).

Thirdly, liquidity is also associated with credit to deposits ratio. It indicates how much a bank lends out of its deposits. A very low ratio indicates that banks are not making full use of their resources. Alternatively, a high ratio indicates more reliance on deposits for lending and a likely pressure on resources. And those banks might not have enough liquidity to cover any unexpected fund requirement (Kumar & Verma, 2008).

Chapter Three Literature Review

Chapter Three Literature Review

3.1 Conceptual Framework

3.1.1 The Concept of Liquidity Risk

There are two sides for the liquidity concept: funding liquidity and market Liquidity. Crockett (2008) defines the market liquidity as the speed to which an asset can be converted into cash easily and without any loss in the asset price. Lou and Sadka (2011) define the liquidity risk as the risk that rises from the security response to changes in market condition. Bakir (2017) and Drehmann and Nikolaou (2013) define the funding liquidity as the ability of an organization or institution to meet or cover its obligation due. Kapadia, Drehmann, Elliott, and Sterne (2012) define it as the ability to pay debt to needed parties.

Liquidity is the major element that is used to pay the expected debts and have liquid assets that are ready to use (Hossain, 2016). When it comes to banking sector, whether they are conventional or Islamic banks, it is simply the ability of the bank to meet its liabilities and finance the investment portfolio, to face any sudden withdraws of its customer deposits.

Each bank should have its adequate level of liquidity and the need to ensure that they pay their contributors when they request their cash (Fiedler, 2000). But it is a truth that cannot be denied that banks are suffering from liquidity risk among other types including interest rate risk and credit rate risk (Duffie & Singleton, 2012).

In general, risk in financial terms is the probability that the actual return may differ from the expected return (Sweeney, 2019). In financial system, there are at least three broad categories of risks, (1) financial risk, (2) business risk, and (3) operational risk. Financial risk concerns risks arising from financial activities of banks, while business risk and operational risk relate to the bank's internal affairs and market conditions. In this respect, liquidity risk is under the financial risk category along with credit risk and market risk.

Diamond, Hu and Rajan (2018) explained that when banks witness sudden withdrawals of customer, this could really cause liquidity shock, especially if the bank was not ready for this situation or have illiquid funds. Zheng (2006) mention that the spread between deposits and loan causes liquidity risk. Cornett, McNutt, Straha and Tehranian (2011) focus on the main reason for liquidity risk, which is high and unexpected, withdraws for deposits, which leads banks to liquid their assets even if that causes loss. Santomero and Babbel (1997) adds that the unstable market condition or market disruption play a crucial role in increasing liquidity risk.

There are many other causes of liquidity risk. Chen, Yang and Yeh (2017) measure the causes of liquidity risk on a sample of 12 countries, and reveal on the strong relationship between liquidity risk and bank performance. They mention the macro factors as influencing factors that

effect on the extent of liquidity risk in addition to regulations and supervision factors.

Cornett (2011) measured liquidity risk during the financial crises and find a relation between liquidity risk and credit risk. Because of the high demand on loans, borrowers were not able to pay back the principles plus the interest. Therefore, banks started to witness high and unexpected levels of defaults that eventually cause some major banks to declare bankruptcy and others to a merger.

It is fair to say that liquidity risk is a major determinant of other risks such as credit risk (Jaara, Jaara, Shamieh, & Fendi, 2017; Yuwonoputro & Syaichu, 2019). Credit risk is the probability of default by the bank borrower to meet obligations according with specific terms. As indicated by Aosaki (2013), the objective of credit risk administration is to augment a bank's hazard balanced pace of return by keeping up credit risk presentation inside adequate parameters.

Basel Committee on Banking Supervision Risk (2008) defines liquidity risk as the variability in the banking ability to fund increases in assets and cover its commitments. It also defines it as a risk resulted when financial associations' securities can't be traded quickly in the market (Kou, Peng & Heyde, 2013).

According to Basel Committee, there are two measures with regard to liquidity risk measurement. The first one is leverage ratio that measures the risk based capital requirements to restrict the usage of leverage. The capital measure for overage called tier 1. Yaacob, Rahman and Karim (2016) focus on the determinant of liquidity risk by using Basel indicators and focused on Islamic Banks in Malaysia. They found out that liquidity risk is affected by CAR and financing and macro factors like GDP that have also relation with liquidity on short and long run.

The study of Hidayat, Al-Khalifa and Aryasantana (2012) looks at the determinants of liquidity risk in the Islamic sector between the period 2004 and 2012 for a sample of 60 Islamic banks. The study reveals that using Islamic financing methods especially Musharakah and Mudarabah can reduce exposure to liquidity risk because they are based on the Islamic principles of profit and loss sharing between the Islamic banks and investor.

3.1.2 Islamic Finance Principles

The genuine difference between conventional banking and Islamic banking lies in the practices, principles and the services provided that are utilized in ordinary fund and prohibited under Shariah laws .So it is vital to mention these principles as discussed in the following sections (Rosman, 2009)

1-Charging an interest

Islam considers loaning with premium installments an exploitative practice that favors the money lender to the disservice of the borrower. According to Shariah law, interest is usury (Riba) which is prohibited.

2-Putting Assets into Associations Connected with Confined Activities

Activities such as producing and selling alcohol or pork are denied in Islam. Therefore, placing assets into such activities is prohibited.

3-Speculation

Shariah cautiously denies any sort of speculation or wagering, which is called *maisir*. Therefore, Islamic cash related associations can't be related with contracts where the obligation regarding depends upon future probable event.

4- Uncertainty and Risk (gharar)

The rules of Islamic finance ban participation in contracts with excessive risk and/or uncertainty. The term *gharar* measures the legitimacy of risk or uncertainty in investments. *Gharar* is observed with derivative contracts and short-selling, which are forbidden in Islamic finance.

5-Material Unalterable Quality of the Trade

Each trade must be identified with a genuine basic financial exchange/money related trade.

6-Profit and Loss Sharing

Parties entering into the contracts in Islamic finance, share advantage/losses related with the trade. It's not possible for anyone to benefit by the trade more than the other party.
3.1.3 Modes of Finance in Islamic Banks

Islamic finance has expanded wonderfully around the globe. Today, Islamic banks offer to their client's different modes to put away their cash and fund their tasks. These modes incorporate answers for short, medium, and long-term financing. These unique kinds of financing game plans of Islamic banks that were created to follow Sharia's principles (Ali ,2004):

1-Murabahah

Murabahah is an Islamic contract for a sale where the buyer and seller agree on the markup (profit) or (cost-plus) price for the item(s) being sold. In Islamic banking it has become a term for a marked-up price and deferred payment, a way of financing a good (home, car, business supplies, etc.). Whereby the bank buys the good and resells it to the customer at higher price (informing the customer of the price increase), and offering to take payment in installments or in a lump sum.

2-Mudarabah

Mudarabah is a profit and loss sharing partnership agreement. Where one partner (financier or rab-ul mal) provides the capital to another partner, (labor provider or mudarib), who is responsible for the management and investment of the capital. The profits are shared between the parties according to a pre-agreed ratio.

3-Musharaka (Partnership)

Musharakah is a form of a joint venture where all partners contribute capital and share the profit and loss on a pro-rata basis. The major types of these joint ventures are:

- Diminishing Partnership: This type of venture is commonly used to acquire properties. The bank and investor jointly purchase a property. Subsequently, the bank gradually transfers its portion of the equity in the property to the investor in exchange for payments.
- 2. Permanent Musharkah: such a joint undertaking doesn't have a specific end date and continues functioning as long as the partaking parties agree to continue with assignments. Generally, it is used to subsidize long term endeavors.

4- Ijarah

Ijarah is a financing agreement, where the lessor (who should own the property) leases the property, to the lessee in exchange for a stream of rental, and purchase payments, ending with the transfer of property ownership to the lessee.

Leasing can also be structured in a way that ends by the purchase of the leased asset in a financing contract. That is, ownership of the leased asset is transferred to the lessee at the end of the lease agreement. This transfer of ownership is made through a new contract, in which the leased asset is either given to the lessee as a gift or is sold to him at a nominal price at the end of the lease agreement. According to a decision of the OIC Fiqh Academy, this second transfer of ownership contract should be signed only after termination of the lease term, not an advance promise to affect such a transfer of ownership.

5-Istisna

Istisna is a contract in which a party orders another to manufacture and provide a commodity. The description includes, delivery date, price, and payment date are all set in the contract. According to a decision of the OIC Fiqh Academy, this type of contract is binding, and the payment of price could be deferred.

6-Salam

It is among the fastest-growing Islamic modes of financing. Salam is a sales contract, in which the price is paid in advance at the time of contracting, against delivery of the purchased goods/services at a specified future date. Not every commodity is suitable for a *salam* contract. It is usually applied only to fungible commodities.

3.1.4 Risks Facing Conventional and Islamic banks

Despite this great diversity of Islamic Banks financial models or investments arrangements, banks still face a group of unique risks that affect their operation and decision making. These risks can be classified into two groups: Those that are faced by conventional banks and risks that are unique to Islamic banks as shown in Figure 3.1.



Figure (3.1): Islamic and conventional banking risks overview

Based on Karim and Archer (2013) risks that are in common between conventional banks and Islamic Banks are:

1) Credit risk

Credit risk is the probability of default as a result of borrower's failure to repay the required loan principal and interest. This will cause an interruption in the cash flow model and high collection costs (Elizalde, 2007).

2) Market risk

Market risk is the risk that an investment may face due to fluctuations in the market. It describes how changes in market prices and

interest rates will result in investment losses. Hasan (2003) distinguishes between:

- a. General or deliberate market hazard, brought about by a development in the costs of all market instruments because of full scale factors (for example a change in monetary strategy).
- b. Unsystematic or explicit market chance, which emerges in circumstances where the cost of one instrument moves off the mark with other comparative instruments.

3) Interest rate risk

It is the changes in interest rates that will affect the value of assets or bond. This risk can be reduced by buying bonds that mature at different dates as well as by hedging fixed-income investments with interest rate swaps and other instruments (Cornford, 2005).

4) Foreign exchange risk

Foreign risk /Currency risk explains the losses that incur because of currency fluctuations in international transactions. These fluctuations or changes cause a decrease in the asset's value.

5) Equity price risk

Equity is referred to equity in companies through the purchase of stocks, and does not commonly refer to the risk in paying into real estate or

building equity in properties. This risk is resulted from holding equity in particular /specific type of investment.

6) Commodity price risk

It is the probability of financial losses in buyer's goods through having higher prices than expected therefore, having higher production costs so lower profits to them.

7) Operational risk

It is the risk of value changes due to external events, such as legal risk, natural disasters or internal events, such as: failed internal processes, systems, and people.

8) Liquidity risk

It is a firm failure to meet its momentary obligation commitments. A higher liquidity will be required if the significant portion of the advantage portfolio comprises of long-term loans and deposits has a high fixation nature. A bank will require higher liquidity.

9) Legal risk

One source of operational risk is legal risk. Islamic banks should be able to identify this risk according to the bank's environment. In addition, legal risk may have a crucial effect also on Islamic banks, because of the lack of reliable legal systems and the uncertainty in the interpretation of financial contracts, amongst other factors (Hassan & Dicle, 2005; Izhar, 2010; Sundararajan & Cihak, 2010).

The risks specific to Islamic banks are as follows (Ariffin, Archer, & Karim, 2009):

1- Displaced Conventional risk

It is the risk accrued to Islamic banks due to the market pressure of having to pay a rate of return equivalent to a competitive rate of return. And absorb a portion of losses, which normally would have been borne by investment account holders, to prevent massive withdrawal of funds.

2- Rate of return risk

According to IFSB, it is uncertainty in returns earned on Islamic Bank's assets. Changes in the market rate of return will have an impact on the Islamic financial institution's market value or net income.

3-Shariah non-compliance risk

According to IFSB, it is the risk arising from the Islamic Bank's failure to comply with the Shariah rules and principles determined by the Shariah board.

4-Equity Investment risk

According to IFSB, it's the risk arising from entering into a partnership in general business activity, as describes in the contract and the provider of finance shares the business risk.

Risk management is the general procedure that a financial institution follows to ascertain the nature of risks and to develop the right strategy (Cumming & Hirtle, 2001). The process of risk management will ensure that the correct sequence of procedures of identification, measurement, and managing of the risks has been followed Rosly (2005). Risk management in Islamic banks consists of risk measurement, risk management and risk control (Van, Greuning & Iqbal, 2007).

Islamic banks face many unique risks because of their operations which must be in match with Islamic conduct (Rosman, 2009). One of these unique risks is liquidity risk. It is one of the top priorities of Islamic bank's assets liabilities management. Liquidity risk in banks comes from the mismatching between the supply and demand of funds. The difference between them is called net liquidity position. Therefore, banks should manage these elements carefully to reduce the liquidity risk.

Since Banks are inspired by different motivations to hold a specific measure of liquid balances. It is crucial to explain the concept of liquidity that refers to the ability of the bank to meet up deposit withdrawals, maturing loan request and liabilities without setback. So, banks defend its customers aligned with troubles of liquidity by captivating in financial liabilities. That can be drained on demand, offering dedicated lending services. The arrangement of balance sheets of banks usually illiquid loans is financed by extremely liquid deposits, or the ability to fund increases in assets and meet obligations as they come due is critical to the ongoing viability of banks.

Since there is a strong association between liquidity and solvency of banks, sound liquidity management reduces the probability of banks becoming insolvent thus, reducing the possibilities of bankruptcies and ban runs. Liquidity management is just as important to the Islamic banks as it is to the conventional banks. However, compared to the conventional counterpart, liquidity management for the Islamic banks is unique and even more challenging. Since most of the existing instruments used for liquidity management are interest-based, therefore, not Islamic law compatible. Also, the rationality of bank customers in the conventional sense in which profit motive prevails in any economic transaction, could result in liquidity withdrawal, from the Islamic banks when return in the conventional counterpart is higher (Kassim, Majid, & Yusof, 2009). Besides, the Palestinian Islamic banks have accomplished consistent development regarding resources size, stores and loaning to the private segment.

3.1.5 Liquidity Risk Management Process

Islamic banks face many restrictions in what kind of activities they can pursue, what types of contracts are allowed to make. Islamic banks have limited access to Islamic money market products and prohibit interbank activities that involve interest. As a result, they hold more liquidity levels than conventional banks that affect the profitability negatively (Brown, Hassan, & Skully, 2007). Banks are commercial organizations with a profit seeking mission. But when talking about risk it is completely different than profit or loss. Risk represents the chance that an investment's actual return will differ from the expected return. Businesses including financial institutions engage in risky activities, because they understand the positive relationship between risk and return. When anticipating a higher return, higher risk must be taken. However, some kinds of risks are uncontrollable, while others are unrewarded. Therefore, it is important to focus on managing an institution's exposure to losses or risk and to protect the value of its assets.

Risk management means taking risks within limits to optimize riskadjusted returns. Risk management in banking sector means the rational implementation of a specific plan to deal with potential losses. Banks usually manage those risks as part of their normal operations, risk management process.

Therefore, banks need risk management process. Risk management process is applied for each phase of analysis, economic and regulatory, within the framework Pyle .Also as stated by Banking Committee on Banking Supervision (BCBS), it is a four step process (BCBS, 2002).

- 1- First step: The identification of risk events and categorizing them into market, credit, operational
- 2- Second Step: The assessment of the risks using risk model. Although there are many methods and approaches of risk management, one popular method that is widely used is the Value at Risk (VaR). It

records different types of risks also assigning risk limits to a bank business unit to have effective capital allocation Linsmeier and Pearson (2000), Saita (1999). On the other hand, traditional/ quantification methods, such as gap, duration and simulation are being used to measure other risks.

- 3- Third step: Monitoring & Mitigation. This step is very important that is done by all financial institutions after analyzing the target risks. It is referred to risk control the aim of which is to choose the right mitigation strategy that minimizes overall risk position.
- 4- Fourth step: Reporting and feedback to make the right decision.

These steps are ongoing process (Mohamad, Mohamad, & Samsudin, 2013). There is no single most ideal way for executing a successful risk management process since banks are not all equivalent. Yet, a risk management framework can be flexibly designed and adapted to match different banks.



Figure (3.2): Liquidity risk management process in banks (Masood, 2016)

Islamic banks face unique risks because of their unique form and their modes of financing such as withdrawal risk, reputation risk, Shariah compliance risk and asset price risk. To handle these risks, it is vital to identify the practices of risk management. Liquidity management is now becoming a more complex phenomenon that needs more understanding (Dahiyat, 2016).

Liquidity management refers to elude cash from depositors and then balancing demands for money from lenders and from depositors asking for their money back (Roseman, 2009). Most of Islamic banks scale adequate capital levels. Islamic banks follow Basel III liquidity measure such as net coverage ratio, which results in that Islamic banks need to hold more assets for funding. This ratio is important for banks to go international and improve its transparency (Harzi, 2012).

Some empirical studies highlight that Islamic banks lacks the flexibility of their non-Islamic counterparts when seeking short-term finance due to the restrictions of Islamic laws (Alman, 2012; Rajhi, Hassairi, & Cergam, 2012; Akhtar, Ali, & Sadaqat, 2011; Ismal, 2010a).

Mohamad (2013) and Ariffin (2012) suggest that ideal methods of liquidity management are not available to Islamic banks. Due to the prohibition of interest transactions Hidayat et al. (2012) and Mounira and Anas (2008) advise that Islamic banks should adopt internal control systems over liquidity risk management process, to avoid liquidity problems at the present and in the future also. An essential part of robust liquidity management program is the credibility and effectiveness of the risk assessment process (Alsayed, 2007). The Islamic financial Services Board (IFSB) as the international standard-setting organization has published two references:

- The Guiding Principles of Risk Management for Institutions Offering Only Islamic Financial Services December (2005).
- 2) The Technical Note on Issues in Strengthening Liquidity Management of Institutions Offering Islamic Financial Services. Ismal (2010) in his research on Shariah issues that arise in the management of liquidity risk discovered that Islamic banks need to develop their liquidity risk management environment as a practice of modern banking standards to ensure safe operations and maintaining business operations.

3.2 Hypotheses development

Based on the previous empirical studies, the determinants of liquidity risk and the distinction between the liquidity risk between Islamic and conventional banks are to be examined through the following hypotheses:

H1: Deposits to assets ratio has a positive impact on the liquidity risk of Palestinian banks.

H2: Return on equity has a positive impact on the liquidity risk of Palestinian banks.

H3: Capital adequacy ratio has a negative impact on the liquidity risk of Palestinian banks.

H4: Bank size has a positive effect on the liquidity risk of Palestinian banks.

H5: The level of liquidity risk is expected to be higher for local banks than foreign banks.

H6: The relationship between DA and the liquidity risk depends on the type of bank, whether Islamic or conventional.

H7: The relationship between ROE and the liquidity risk depends on the type of bank, whether Islamic or conventional.

H8: The relationship between bank size and the liquidity risk depends on the type of bank, whether Islamic or conventional.

H9: The relationship between capital adequacy and the liquidity risk depends on the type of bank, whether Islamic or conventional.

3.3 Conceptual frame work

The framework of this study is summarized in Figure 3.3.



Figure (3.3): Conceptual frame work

Chapter Four Research Methodology

Chapter Four Research Methodology

A mixed research methodology will be used in this thesis, the quantitative research methodology and the qualitative research methodology. The quantitative research is used to model the liquidity risk determinants and to examine how the effects of these determinants are different in Islamic banks from conventional banks. Qualitative research is used then to explore the liquidity problems and management practices in the context of Palestine.

4.1 Quantitative research methodology

4.1.1 Data and Sample Descriptions

The data that have been used in this thesis are micro data in nature, which is related to specific banks. For the purpose of construction of the study variables, the financial statements of the banks (conventional and Islamic) are collected and variables are constructed manually. The sources of the data include the published financial reports of the banks available from Association of Banks in Palestine. The study covered the period from 2009 to 2018. We have a full sample of 12 Islamic and conventional banks. For conventional banks; we selected 10 conventional banks whereas 2 Islamic banks have been selected. Due to missing data issue, the data set for Al -Safa bank and Jordan commercial bank were excluded in order to have strongly balanced panel data.

4.1.2 Model Specification, Estimation and Variables Definition

Multiple Linear regressions were conducted to determine the impact of DA, ROE, Size, Nationality and CAR on the liquidity risk of banking sector in Palestine and the effect of bank type on these relationships.

After conducting OLS regression analysis and since the presence of serial correlation in our regression model that made the coefficients inefficient, and because one of the assumption of the error terms that have a constant variance was violated. Therefore, the generalized least square method (GLS) method has been conducted instead of OLS to improve our model and eliminate the heteroskedasticity problem.

Another model was used to examine the effect of type of bank on the relationship between factors affecting liquidity and the liquidity risk in Palestinian banks.

In our research two models were used: The first model contained the main independent variables. The second model contained an interaction variable (interaction effect of Islamic banks) to investigate whether independent variables were affecting Islamic banks differently from conventional banks.

Model one:

 $LRit=\alpha 0+\alpha 1.DAit+\alpha 2.ROEit+\alpha 3.BSit+\alpha 4.CARit+\alpha 5.D1+\alpha 6.D2+\varepsilon it$

Model two:

$LRit = \alpha 0 + \alpha 1.DAit + \alpha 2.ROEit + \alpha 3.BSit + \alpha 4.CARit + \alpha 5.D1 + \alpha 6.D2 + \alpha 7.$

D1*DA+ α 8.D1*ROE+ α 9.D1*BS+ α 10.D1*CAR + ε it

Whereas:

 $\alpha 0$ to $\alpha 10$: regression coefficients

LRit: liquidity risk.

DAit: deposits to assets ratio.

ROEit: return on equity.

BSit: bank size.

CARit: capital adequacy ratio.

D1: Dummy variable equal to 1 if the bank is Islamic.

D2: Dummy variable equal to 1 if the bank is local.

Eit: error term.

4.1.3 Variables Measurement

Table 4.1 illustrates the variables' operational definitions that have been employed in our investigation.

Variable	Abbaar	Maggungant			
variable	Abbrev.	Measurement			
1.Dependent Variable: Liquidity Risk					
Liquidity Dick	LR	Total customer deposits/Liquid assets			
Elquidity Kisk		(Mennawi and Ahmad,2020)			
2. Independent and moderator Variables					
Deposits to	D٨	Total doposite/Total assots (Aby Zir 2016)			
Assets	DA	1 otal ueposits/1 otal assets(Abu Zir, 2016)			
	ROE	Earnings available for common stockholders/Common stock equity (Bagh,			
Return on Equity					
		Razzaq, Azad, Liaqat and Khan,2017).			
Capital adequacy		Tier1+Tier2/risk weighted assets ((Laeven,			
ratio	CAK	Ratnovski and & Tong, 2014)			
Bank size	BS	Logarithm of assets			
Islamic	D1	Dummy variable equal to 1 if the bank is			
	DI	Islamic, zero otherwise			
Local	D2	Dummy variable equal to 1 if the bank is			
Local	D2	local, zero otherwise			

 Table (4.1): Measurement of Variables

Discussion of variables followed.

1. Deposit to Assets ratio (DA): it is being measured as total deposits to total assets. Deposit to assets ratio is considered an efficiency measure that examines the bank's reliance on deposit funding. Deposits to asset ratio measures the magnitude of assets being funded by total deposits.

The effect of this variable was tested by Cucinelli (2013) and Alzoubi, (2017). Also used by Otieno et al. (2016) who conducted his study by using total deposit to total asset ratio. These studies found that there is a positive relation between deposits to assets and bank liquidity risk.

2. Return on equity (ROE): it is being measured as earning available for common stockholders to common stock equity. Return to equity is considering a profitability measure, as it measures a company's ability

to earn a return on its equity investments. ROE may increase without additional equity investments, as the ratio can rise due to higher net income due to a larger asset base funded with debt (Laeven, Ratnovski & Tong, 2014).

The effect of this variable was tested by Wanjohi and Xiaodong (2011). The finding of their study showed that ROE seemed to have a positive and relationship with liquidity risk.

3. Capital adequacy ratio (CAR): it is being measured as tier 1 and tier 2 capital to risk weighted assets. Capital adequacy ratio is considered a solvency ratio that provides way to evaluate a company's debt versus its revenues situation. However, the capital adequacy ratio is usually applied specifically to evaluate banks. The reason capital adequacy ratios (CAR) is critical is to make sure that banks have enough cash to absorb a reasonable amount of losses before they become insolvent and consequently lose depositors' funds. The capital adequacy ratios ensure the efficiency and stability of financial system by lowering the risk of banks becoming insolvent. Generally, a bank with a high capital adequacy ratio is considered safe and likely to meet its financial obligations (Laeven, Ratnovski & Tong, 2014).

The effect of this variable was tested by Staroselskaja (2014) who found that there is a negative relationship between CAR and liquidity risk.

4. Bank size (BS): it is being measured as the logarithm of the banks total assets. Increasing bank size can increase profitability by allowing banks

to realize economies of scale. For example, increasing size allows banks to spread fixed costs over a greater asset base, thereby reducing their average costs. Furthermore, as the scale of operation increases, banks could better use specialized inputs such as loan officers with expertise in a particular business line, resulting in greater efficiency (Laeven, Ratnovski & Tong, 2014).

5. Islamic banks

A dummy variable equal to 1 if the bank is Islamic, zero otherwise.

6. Local Banks

A dummy variable equal to 1 if the bank is local, zero otherwise.

4.2 Qualitative Research Methodology

A qualitative research methodology has been used through an openend questionnaire distributed among the target sample of Islamic and conventional bankers. These questionnaires were answered by the managers of Treasury and Risk department for both Islamic and conventional banks.

4.2.1 Data and sample description

The sample chosen to explore qualitatively the issues related to liquidity management in Palestine. Sample includes six banks (three conventional and three Islamic) as in Table 4.2.

Islamic banks	conventional banks
Arab Islamic bank	Bank of Palestine
Palestine Islamic bank	Palestine Investment Bank
Al-Safa bank	Al-Quds Bank

Table (4.2): Sample Description

The reason for choosing the above three conventional banks besides the time and approvals limitations is that these banks are local Palestinian banks therefore their policies and procedures are not affected by external headquarter influences.

4.2.2 Questionnaire Design

The questionnaires (Annex 1 and Annex 2) that were directed to Islamic bankers (treasury managers and risk department) and conventional bankers aimed to answer the following questions:

- Defining the differences in liquidity risk measurement and management between Islamic and conventional banks.
- 2. Exploring the types of liquidity risk in Palestinian Islamic and conventional banks.

4.3 Propositions

Depending on the research questions this research aims to explore the following propositions:

Proposition 1:

Islamic Banks faces a unique liquidity risk because of its nature of asset-based financing.

Proposition 2:

Islamic and conventional Palestinian Banks use different liquidity risk management techniques to effectively reduce liquidity risk problems

Proposition 3:

Islamic and conventional banks don't have dedicated committee to oversee the liquidity risks.

Chapter Five Results and Discussion

Chapter Five Results and Discussion

5.1 Preface

This thesis used two approaches to answer the questions of the study: quantitative and qualitative approach.

5.2 Quantitative approach

This approach was used to model the factors that affect liquidity risk and to investigate the hypotheses of the thesis.

5.2.1 Summary Statistics

The summary statistics of the study variables for conventional and Islamic banks are presented below in Table 5.1. The table contains the mean, median, maximum value, minimum value, standard deviation, for selected variables (dependent and independent). This table gives a quick and simple description of the data.

	Mean	Median	Maximum	Minimum	Std. Dev.	Observations
LR	0.755	0.766	6.231	0.110	0.498	133
DA	0.764	0.833	0.901	0.033	0.160	133
ROE	-0.058	0.070	0.379	-10.300	1.018	133
BS	20.097	20.092	22.309	18.282	0.944	133
CAR	0.360	0.242	1.289	0.111	0.282	133
ISLAMIC	0.211	0.000	1.000	0.000	0.409	133
LOCAL	0.549	1.000	1.000	0.000	0.499	133

Table (5.1): Summary statistics for Islamic and conventional banks

The mean value of LR is 0.755; this shows that percentage of LR is 70 % which is high while the standard deviation is 0.49. The mean value of

deposit to asset ratio (DA) is 0.76, which is also high, while the (DA) registered a low standard deviation of 0.160. The mean value of ROE is – 5% which is a warning sign, but the median is 7% indicating that the mean is influenced by outlier values. The standard deviation for ROE shows that the percentage is more than 100% which is very high. In regard to capital adequacy ratio (CAR), the mean value is 0.36 which around 30%, while the standard deviation is 0.49.

By the exception of bank size, the mean-median ratio is approximately 1, which indicates the normality of data. Compared to mean standard, values of error are low which represents small coefficient of variation and indicates low spread in the data.

5.2.2 Correlation analysis

Correlation analysis is utilized to figure out the strength of the relation among variables. This relation can happen among independent variables and dependent ones, as well as it can be among independent variables facing other independent variables. Generally, the high correlation between independent variables shows the existence of multicollinearity.

In this examination, we have tested correlations in Islamic and conventional Banks in one group. Table 5.2 illustrates that the relationship of LR ratio is positive with DA ratio, ROE and bank size, negative with CAR.

	LR	DA	ROE	BS	CAR	ISLAMIC	LOCAL
LR	1.000						
DA	0.158	1.000					
ROE	0.079	0.161	1.000				
BS	0.154	0.539	0.221	1.000			
CAR	-0.064	-0.504	-0.222	-0.584	1.000		
ISLAMIC	-0.087	-0.164	0.077	-0.218	-0.133	1.000	
LOCAL	0.104	0.158	0.032	0.134	-0.307	0.283	1.000

 Table (5.2): Full sample correlation matrix

According to the correlation Table 5.2, there are no strong relationships between the independent variable. Since that the multicollinearity problem means that there are at least two independent variables which have a strong relationship between each other. As we don't have that strong relationship 70% or more between two independent variables, we can conclude that we don't have a multicollinearity problem in our model.

5.2.3 Model Estimation

Two models have been estimated using the panel data regression under GLS as viewed in Table 5.3. Summary of the regression results for Model 1 and Model 2 are shown in Table 5.3.

Dependent Variable: LR	Model 1			Model 2			
Variable	Coefficient	t-Statistic	Prob.	Coefficient	t-Statistic	Prob.	
DA	0.1568	2.4107	0.0174	0.269	2.535	0.013	
ROE	0.0065	0.6607	0.5100	0.005	0.288	0.774	
BS	0.0438	4.1379	0.0001	0.039	2.850	0.005	
CAR	-0.0890	-2.8203	0.0056	-0.037	-0.742	0.460	
ISLAMIC	-0.0564	-2.8170	0.0056	0.384	0.693	0.490	
LOCAL	0.0659	5.2121	0.0000	0.050	2.842	0.005	
DA*ISLAMIC				-0.254	-2.001	0.048	
ROE*ISLAMIC				0.057	2.251	0.026	
BS*ISLAMIC				-0.006	-0.221	0.826	
CAR*ISLAMIC				-0.258	-3.602	0.001	
С	-0.2332	-1.1395	0.2566	-0.248	-0.949	0.345	
R-squared	0.582227			0.629			
Adjusted R-squared	0.562334			0.599			
F-statistic	29.26659			20.727			
Prob(F-statistic)	0.000			0.000			
Total panel (unbalanced) observations	13+3			133			
Method: Panel EGLS (Cross-section weights)							

Table (5.3): Estimation results for liquidity risk

The regression results of model 1 (which represents the effect of DA, ROE, size, nationality and CAR on liquidity risk management of banks in general) and model 2 (which adds the interaction effect of Islamic banks dummy with each of the independent variables to compare Islamic with conventional banks). Model 1 and model 2 have generated the values of R squared of 58% and 62% respectively. However, the better measurement of the explanatory power in an econometric model is the adjusted R squared, as this value is adjusted for the number of explanatory variables, and prevents the random increase in R squared. The value of adjusted R squared for the first model is 56%, which indicates that 56% of the variance in the dependent variable (LR) is explained by the variability in

the examined independent variables.

On the other hand, the value of adjusted R squared for the second model which has been tested with the incremental effect of the moderating variable "Islamic Bank" is 59%, which means that only 59 percent of the variance in the dependent variable (LR) is attributed to the variability of the independent variables and the interaction variable. Obviously, both models are close to each other and the explanatory powers for both models are good.

In regards to the overall significance of both regression models, this can be measured by F-test. According to this test, the regression analysis reveals that both models are statistically significant at the empirical significance level of 1% each, as the results suggested probability values of F-test of almost zero for both models

Given that the empirical significance level of this study is 5%, the GLS regression test revealed that the following variables: DA (deposit to asset ratio), bank size, nationality and type of the bank and CAR (Capital adequacy ratio) from the first model are significant. In the second model, DA, size, nationality and the ffect of interaction variable "Islamic bank" with DA, ROE and CAR from the second model were found to be statistically significant.

5.2.4 Discussion of Estimation Results for Model One

The regression results for the Model 1 reveals that the explanatory variable DA (deposit to asset) has a significant positive relation with LR

(liquidity risk) in banks in general, with the coefficient value of 0.1568. This indicates that one unit of increase in DA or the ratio of total deposit to total asset increases the ratio of liquidity risk by 0.1568 units. This outcome is consistent with Cucinelli (2013) and Alzoubi, (2017). Also, the same relation was found by Otieno et al. (2016) who conducted his study by using total deposit to total asset ratio, he found a similar positive relation with bank liquidity risk.

As stated by Model one in terms of Deposits to assets (DA) which has significant positive relation with LR (liquidity risk), the null hypothesis will be rejected and alternative hypothesis will be accepted since p-value 1% is less than significant level 5%. The Deposits to assets ratio has a positive impact on the liquidity risk of Palestinian banks.

If the management of banks will rely more on their deposits to fund their assets that might incur more exposure to liquidity risk and related to the sudden withdraw. The higher the ratio, the more the bank must rely on external funding, which is often a more costly source of funding than deposits.

As it can be seen from Table 5.3, the negative significant determinant of the liquidity risk in Palestine is the ratio of capital adequacy CAR. This ratio has a negative coefficient of 0.0890 and has a significant level of 0%. Which indicates that an increase of one unit in this ratio will cause the liquidity risk LR (The ratio of customer deposits to liquid assets) to be decreased by 0.0890 units. Since a higher value of this ratio reflects a

more capital a bank retains compared to its risk, this means that a higher a bank's CAR, the more likely it is to be able to withstand a financial downturn and less likely exposure to liquidity risk. The same conclusion was found by Staroselskaja et al. (2014) who attempted to identify the determinants of liquidity risk and examined the relation between capital adequacy ratio and liquidity risk, and found a negative relationship of CAR with liquidity risk.

As stated by Model one in terms of capital adequacy ratio (CAR) which has significant negative relation with LR (liquidity risk), the null hypothesis will be rejected and alternative hypothesis will be accepted since p-value 0% is less than significant level 5%.

The banks' management has to keep a high capital adequacy ratio to ensure efficiency and stability of their financial system, and to stay safe and have less likely to exposure of liquidity risk.

Another positive but insignificant determinant of the liquidity risk in Palestine is the ratio of return on equity which is an indicator of bank performance. This result is consistent with Wanjohi the finding of their study showed that ROE seemed to have a positive and insignificant relationship with liquidity risk.

As stated by model one in terms of Return on equity (ROE) which has insignificant relation with LR (liquidity risk), the null hypothesis will be accepted and alternative hypothesis will be rejected since p-value 51% is higher than significant level 5%. In addition to that, in model the effects of Islamic and local banks from the first model are as follows: Islamic banks: The effect of Islamic banks has negative significant effect on liquidity risk (coefficient is -5%, p value is 0%) whereas the effect of local banks has positive significant on liquidity risk according to coefficient 6% and p-value of 0%. These results mean that Islamic banks have lower level of liquidity risk compared to conventional banks while local banks have higher levels of liquidity risk compared to foreign banks.

As illustrated by previous studies, conventional banks are better than Islamic banks in managing their liquidity risk. The alternative hypothesis will be accepted. So that Islamic banks are better than conventional banks in managing their liquidity risks. Foreign banks are better than local banks in managing liquidity risk.

5.2.5 Discussions of Estimation Results for Model Two

The test for the second model was implemented with the interaction variable (Islamic bank dummy) and as it can be seen from the Table 5.3, this analysis shows that the independent variable DA (deposit to assets), bank size, and bank nationality has positive effects on liquidity risk. The interaction of DA with Islamic bank dummy has a negative significant relations with the dependent variable liquidity risk in opposite with the DA variable relationship. According to coefficient -.254, the deposits to assets ratio has weaker effect in Islamic banks on liquidity risk than conventional banks.

The incremental effect of the interaction variable (Islamic bank) on the DA in this model, has mitigate the effect of deposit to asset on the liquidity risk, the interaction variable has reduced the power of relationship between DA on the liquidity risk. This indicates that one unit of increase in the ratio of total deposit to total asset with the effect of Islamic bank will lower than conventional banks by 0.254 units. This outcome is consistent with Leykun (2016), the results of his study revealed that the total deposit to total asset ratio affects the liquidity risk of banks negatively. This supports that Islamic banks have better liquidity position in regard to deposit-base and diversity. As stated by model two in terms of interaction with deposits to assets (DA), the null hypothesis will be rejected and alternative hypothesis will be accepted since p-value 4% is lower than significant level 5%.

Other independent variables ROE and CAR were also significant. Due to the incremental effect of Islamic bank on ROE in the model, this lead to significantly positively relationship between interaction variable with ROE and liquidity risk. This indicates that one unit of increase in ROE for Islamic bank will increase the liquidity risk by 0.057 units more than the effect on conventional banks. This finding concurs with that of Oino (2014) who posted a positive significant relationship between ROE and liquidity risk for banks in Ghana. The liquidity risk may increases while the ROE increase due to the increase in income. This is in line with the research of Dahiyat (2016) which concludes that there is an unfavorable influence between liquidity risk and profitability. As stated by model two in terms of return on equity (ROE), the null hypothesis will be rejected and alternative hypothesis will be accepted since p-value 2% is lower than significant level 5%.

The incremental effect of the interaction variable (Islamic bank) on the capital adequacy ratio CAR in this model has emphasis the negative direction of the relationship between capital adequacy ratio and the liquidity risk. This indicates that one unit of increase in CAR with the effect of Islamic bank will decreases the liquidity risk by 0.258 units in Islamic banks more than conventional banks. This result consistent with Duqi and Al-Tamimi (2018), as they concluded a negative and significant relationship between the capital adequacy ratio and liquidity risk.

As stated by model two in terms of capital adequacy ratio (CAR), the null hypothesis will be rejected and alternative hypothesis will be accepted since p-value 0% is lower than significant level 5%.

To avoid the exposure to liquidity risk, banks' management should keep a capital requirement. This ratio represents the likelihood of insolvency in banks. In other words, sufficient capital ratio maintains a stable financial system and hedges bank from operation risk, market risk, and liquidity risk.

5.4 Qualitative Results and Discussions

This part provides discussion of the finding of the open-ended questionnaire that conducted with Islamic and conventional bankers (treasury managers and risk department). In this study, an open-end questionnaire has been written, developed and data were collected from the Islamic and conventional bankers to understand the needed aspects of those parties and to answer the research questions.

Measurement of liquidity risk

- 1- Conventional and Islamic banks use a daily report called the "Report of the Statement of Cash" in branches. This report is prepared by the Risk department of and summarized by measuring the gaps between the cash inflows and cash outflows.
- 2- A detailed statement of the amount of deposits in banks and all branches are also prepared.

The nature of liquidity risk in Palestinian Islamic and Conventional banks

For Islamic banks:

- 1- The main source is customer deposits, which are short-term and the bank is unable to benefit from these deposits with them due to the possibility of sudden withdrawal.
- 2- When customers deposit certain amounts of cash in terms of Hawala or checks, a cash of the same value must exist inside the treasury. The bank doesn't benefit from that cash because it doesn't own it.

For conventional banks these risks are less noticeable.
Existence of risk management committee specialized with the liquidity risk

Islamic banks

All Islamic banks have a special division called Asset and Liability Management Committee (ALCO); Islamic banks coordinate with Shariah Supervisory Board for liquidity risk management and follow directions of Palestine Monetary Authority management.

The General Manager of the Bank is responsible for ALCO committee and a member of it, the committee members are general manager, treasury manager, risk manager and finance department and asset development manager.

ALCO committee has the following duties:

- 1. Assets and Liabilities Management
- 2. Deposits and Finances Management
- 3. Branch Plans for the Bank
- 4. Investments
- 5. Rate of Returns
- 6. Other Banks Offers
- 7. Cash Flow in Treasury
- 8. Cash Shipment and Cash in Branches.

Conventional banks

Conventional banks have different responses with this matter as following:

1- Bank of Palestine: has executive committee that is responsible for dealing with liquidity issues. The head of this committee is the general manager and the members are the members of the executive committee, this committee meets at least 12 times yearly, when a specific emergency or problem occurs, it holds an immediate, unscheduled meeting.

The roles of this committee include facilities granting, deposits, finances management, branch plans for the bank, investments, and rate of returns.

- 2- Al-Quds bank has ALCO committee that is responsible for dealing with liquidity issues. The general manager of the bank is responsible for this committee and a member of it, the committee consists of the general manager, treasury manager, risk manager, finance department and asset development manager meeting every 2 months. The role of this committee is continue management of day to day transactions, long-term work, deposits and cash inflows and outflows currency exchange rated.
- 3- Bank of Investment has a treasury department. The head of this committee is the treasury manager, meets daily for urgent cases but regularly meets monthly. The role of this committee is facilities granting, deposits, finances management, branch plans for the bank, investments, and rate of returns.

Chapter Six

Recommendations and Future Research Directions

Chapter Six

Recommendations and Future Research Directions

The recommendations of the study are divided into two parts. The first part provides recommendations to Islamic and conventional banks. The second part provides recommendations to regulators of Islamic and conventional banks in Palestine, namely Palestine Monetary Authority (PMA).

The following are specific recommendations for Islamic and conventional banks:

- 1- The management of conventional and Islamic banks has to manage capital adequacy ratio to ensure efficiency and stability of their financial system, and to stay safe and have less likely to exposure of liquidity risk.
- 2- The liquidity management process in both categories of banks (Islamic and conventional) is quite informal, where cash requirements are anticipated based on experience. The operations manager of banks should be better trained for effective liquidity management in Islamic banks in Palestine.
- 3- An independent liquidity office should be placed for conventional and Islamic banks. Because banker agrees on the existence of special committee to deal with a lot of issues including liquidity but under a different title, so the independent office should be united, and have its major role in dealing only with liquidity issues, practices method, and

implementations. This recommendation is presented after conducting the questionnaire with Islamic and conventional banks.

- 4- Banks should increase their investment portfolio because the presence of the liquidity problem with banks means not to fully employ the sources of funds, banks, especially Islamic ones are paying a suboptimal return for investors, so Islamic banks are required to be more diversified in its portfolio about the new financing forms of Murabaha and Musharaka.
- 5- Establishing a financial liquidity management center as a way for Islamic financial institutions to manage liquidity through liquid investments of short or medium term and legally acceptable.

Lastly, banks regulators are also provided with the following recommendation:

- Palestine Monetary authority should stand ready to help and bailout any bank facing liquidity-related issues and problems. This recommendation was based on the questionnaire results.
- 2- Palestine Monetary authority should construct some other standard which is consistent with norms for better utilization and to meet the concept of Islamic and conventional banking in Palestine.
- 3- Palestinian Monetary Authority must review the nature of the current liquidity relations with Israeli banks and find a new sustainable mechanism that serves the interests of the Palestinian banking sector

References

- Abbadi, S. M., & Karsh, S. M. A. (2013). Methods of evaluating credit risk used by commercial banks in Palestine. International Research Journal of Finance and Economics, 111, 146-159.
- Abdelkarim, N., & Burbar, M. (2017). How Banks in Palestine Manage
 Financial Risk?. Remah Review for Research and Studies, 21(4227),
 1-38.Abu Zir, M. (2016). Determinants of banks' profitability: lessons
 from Palestinian banks (Doctoral dissertation).
- Abugamea, G. (2018). Determinants of Banking Sector Profitability: Empirical Evidence from Palestine (No. 89772). University Library of Munich, Germany.
- Ahlasa & Ramadan (2013). *The role of accounting information in managing liquidity risk*, An applied study on commercial banks operating in the Gaza Strip(Master thesis,Islamic university ,Gaza strip).Retrieved from:<u>http://hdl.handle.net/20.500.12358/18261</u>
- Akhtar, M. F., Ali, K., & Sadaqat, S. (2011). Liquidity risk management: a comparative study between conventional and Islamic banks of Pakistan. Interdisciplinary journal of research in business, 1(1), 35-44.
- Al-Arabid, Kh. M. s. Kh. (2012). Analytical study of factors affecting profitability in Palestinian commercial banks with the aim of improving performance: a field study.(Master thesis, Cairo

University, Cairo). Retreived from: <u>https://iugspace.iugaza.edu.ps/bitstre</u> am/handle/20.500.12358/18398/file 1.pdf?sequence=1&isAllowed=y

- Al-Darraji, Hassan Ahmed, & Hamidi, Youssef Mamdou. (2015). The Impact of Risk Management on the Degree of Safety in the Libyan Banking System: An Empirical Study on Public Commercial Banks Operating in Libya. al-Majallah al-• Arabīyah lil-• Ulūm al-Ijtimā• īyah, 117 (3938), 1-21
- Ali, S. S. (2004, February). Islamic modes of finance and associated liquidity risks. In conference on Monetary Sector in Iran: Structure, Performance & Challenging Issues, February, Tehran, Iran, 21(1).
- Alman, M. (2012). Shari'ah supervisory board composition effects on Islamic banks' risk-taking behavior. Journal of Banking Regulation, 14(1), 134-163..
- Alsayed, A. (2007). Risk Management Issues in Islamic Banking. *Islamic Finance Review, Euromoney Year Books*, 8, p20-24.
- Alzoubi, T. (2017). Determinants of liquidity risk in Islamic banks. *Banks & bank systems*, (12,№ 3), 142-148.
- Aosaki, M. (2013). Implementation of Basel III: Economic Impacts and Policy Challenges in the United States, Japan, and the European Union. Walter H. Shorenstein Asia-Pacific Research Center, Freeman Spogli Institute for International Studies, Stanford University,2(16)

- Archer, S., & Karim, R. A. A. (2013). Supervision of Islamic Banks: The Regulatory Challenge—Basel II and Basel III. Islamic Finance: The New Regulatory Challenge, 1.
- Arif, A., & Anees, A. N. (2012). Liquidity risk and performance of banking system. *Journal of Financial Regulation and Compliance*, 20(2), 182-195.
- Ariffin, N. M.(2012). Liquidity Risk Management and Financial Performance in Malaysia: Empirical Evidence From Islamic Banks. Aceh International Journal of Social Sciences, 1(2), 26309.
- Ariffin, N. M., Archer, S., & Karim, R. A. A. (2009). Risks in Islamic banks: Evidence from empirical research. *Journal of Banking Regulation*, 10(2), 153-163.
- Association of Banks in Palestine. (2014). Financial Position Report 2014. Retrievedfrom<u>https://www.abp.ps/files/server/Financial%20Position%</u> 202014.pdf
- Bagh, T., Razzaq, S., Azad, T., Liaqat, I., & Khan, M. A. (2017). The causative impact of liquidity management on profitability of banks in Pakistan: An empirical investigation. *International Journal of Academic Research in Economics and Management Sciences*, 6(3), 153-170.
- Bakir, C. (2017). How can interactions among interdependent structures, institutions, and agents inform financial stability? What we have

still to learn from global financial crisis. *Policy Sciences*, 50(2), 217-239.

- Barnes, P. (1987). The analysis and use of financial ratios: A review article. *Journal of Business Finance & Accounting*, *14*(4), 449-461.
- Basel Committee on Banking Supervision. (2002). Sound Practices for the Management and Supervision of Operational Risk. Retrieved from https://www.bis.org/publ/bcbs91.htm
- Bayyoud, M., & Sayyad, N. A. (2015). The impact of internal control and risk management on banks in Palestine. *International Journal of Economics, Finance and Management Sciences*, 3(3), 156-161.
- Brown, K. E., Hassan, M. K., & Skully, M. T. (2007). Operational efficiency and performance of Islamic banks. In Handbook of Islamic Banking (pp. 96-115). Edward Elgar Publishing.
- Chen, R. R., Yang, T. H., & Yeh, S. K. (2017). The liquidity impact on firm values: The evidence of Taiwan's banking industry. *Journal of Banking & Finance*, 82, 191-202.
- Čihák, M., & Hesse, H. (2010). Islamic banks and financial stability: An empirical analysis. Journal of Financial Services Research, 38(2-3), 95-113.
- Cornett, M. M., McNutt, J. J., Strahan, P. E., & Tehranian, H. (2011). Liquidity risk management and credit supply in the financial crisis. *Journal of financial economics*, *101*(2), 297-312.

- Cornford, A. (2005). Basel II: the revised framework of June 2004 (No. 178). United Nations Conference on Trade and Development.
- Crockett, A. (2008). Market liquidity and financial stability. Financial Stability Review, (11), 13-17.
- Cucinelli, D. (2013). The determinants of bank liquidity risk within the context of euro area. *Interdisciplinary Journal of Research in Business*, 2(10), 51-64.
- Cumming, C., & Hirtle, B. (2001). The challenges of risk management in diversified financial companies. *Economic Policy Review*, 7(1).
- Dahiyat, A. (2016). Does liquidity and solvency affect banks profitability?
 Evidence from listed banks in Jordan. *International Journal of Academic Research in Accounting, Finance and Management Sciences*, 6(1), 35-40.
- Diamond, D. W., Hu, Y., & Rajan, R. G. (2018). The Spillovers from Easy Liquidity and the Implications for Multilateralism. *IMF Economic Review*, 1-31.
- Drehmann, M., & Nikolaou, K. (2013). Funding liquidity risk: definition and measurement. *Journal of Banking & Finance*, *37*(7), 2173-2182.
- Duffie, D., & Singleton, K. J. (2012). Book of Credit risk: pricing, measurement, and management. Princeton university press ,26(1), 60-285.

- Duqi, A., & Al-Tamimi, H. A. H. (2018). The impact of owner's identity on banks' capital adequacy and liquidity risk. *Emerging Markets Finance and Trade*, 54(2), 468-488.
- Durrah, O., Rahman, A. A. A., Jamil, S. A., & Ghafeer, N. A. (2016). Exploring the relationship between liquidity ratios and indicators of financial performance: An analytical study on food industrial companies listed in Amman Bursa. International Journal of Economics and Financial Issues, 6(2).
- Elizalde, A. (2007). From Basel I to Basel II: an analysis of the three pillars. *Documentos de Trabajo (CEMFI)*, (4), 1.
- Fiedler, R. E. (2000). Liquidity risk. The Professional's Handbook of Financial Risk Management. Oxford: Butterworth-Heinemann,2(2) 441-472.
- Harvey, J., & Spong, K. (2001). The decline in core deposits: what can banks do?. *Financial Industry Perspectives*, (dec),35-48.
- Harzi, A. (2012). The Impact of Basel III on Islamic Banks: A Theoretical Study and Comparison with Conventional Banks اثر بازل 3 على المصارف التقليدية. Chapters of books published by the Islamic Economics Institute, KAAU or its faculty members., 591-610.
- Hasan, I., & Marton, K. (2003). Development and efficiency of the banking sector in a transitional economy: Hungarian experience. Journal of Banking & Finance, 27(12), 2249-2271.

- Hassan, M. K., & Dicle, M. F. (2005). Basel II and regulatory framework for Islamic banks. *Journal of Islamic Economics, Banking and Finance*, 1(1), 1-16.
- Hidayat, S. E., Al-Khalifa, M. D., & Aryasantana, A. G. P. (2012). A survey on the level of effectiveness of liquidity risk management of Islamic banks in Bahrain. International Research Journal of Finance and Economics, 91, 39-45.
- Hossain, I., & Choudhury, M. A. (2016). Conclusion: the Islamization of knowledge and its implications. Islamic Financial Economy and Islamic Banking, Taylor and Francis, London, 252-256.
- Ibe, S. O. (2013). The impact of liquidity management on the profitability of banks in Nigeria. Journal of Finance and Bank Management, 1(1), 37-48.
- Islamic Financial Services Board. (2019). Islamic Financial Services Stability Report. Retrieved from <u>http://www.ifsb.org/sec03.php</u>

Ismal, R. (2010). Managing banking liquidity risk in the current economic conditions: A conceptual framework. *Journal of Management & Public Policy*, *1*(2), 48-63.

Ismal, R. (2010a). How do Islamic banks manage liquidity risk? An empirical survey on the Indonesian Islamic banking industry. Kyoto Bulletin of Islamic Area Studies, 3(2), 54-81.

- Izhar, H. (2010). Identifying operational risk exposures in Islamic banking. *Kyoto Bulletin of Islamic Area Studies*, *3*(2), 17-53.
- Jaara, O. O., Jaara, B. O., Shamieh, J., & Fendi, U. A. (2017). Liquidity Risk Exposure in Islamic and Conventional Banks. *International Journal of Economics and Financial Issues*, 7(6), 16.
- Jedidia, K. B., & Hamza, H. (2015). Determinants of liquidity risk in Islamic banks: A panel study. *Islamic Management and Business*, 2(2), 137-146.
- Kapadia, S., Drehmann, M., Elliott, J., & Sterne, G. (2012). Liquidity risk, cash flow constraints, and systemic feedbacks. In *Quantifying systemic risk* (pp. 29-61). University of Chicago Press.
- Kassim, S. H., Majid, M. S. A., & Yusof, R. M. (2009). Impact of monetary policy shocks on the conventional and Islamic banks in a dual banking system: Evidence from Malaysia. *Journal of Economic Cooperation and Development*, 30(1), 41-58.
- Keramati, M. A., & Shaeri, M. (2014, July). Assessment of Credit Risk Management and Managerial Efficiency of Banks Using Data Envelopment Analysis (DEA) Network. In *Biological Forum* (Vol. 6, No. 2, p. 320). Research Trend.
- Kou, S., Peng, X., & Heyde, C. C. (2013). External risk measures and Basel accords. *Mathematics of Operations Research*, 38(3), 393-417.

- Kumar, N., & Verma, P. (2008). Credit deposit ratio and ownership structure in the Indian banking sector: an empirical analysis. *Global Academic Society Journal: Social Science Insight*, 1(4), 4-17.
- Laeven, L., Ratnovski, L., & Tong, H. (2016). Bank size, capital, and systemic risk: Some international evidence. Journal of Banking & Finance, 69, S25-S34.
- Lartey, V. C., Antwi, S., & Boadi, E. K. (2013). The Relationship between Liquidity and Profitability of Listed Banks in Ghana. International Journal of Business and Social Science, 4(3).
- Leykun, F. (2016). Determinants of commercial banks' liquidity risk: Evidence from Ethiopia. *Research Journal of Finance and Accounting*, 7(15), 47-61.
- Linsmeier, T. J., & Pearson, N. D. (2000). Value at risk. *Financial Analysts Journal*, 56(2), 47-67.
- Lou, X., & Sadka, R. (2011). Liquidity level or liquidity risk? Evidence from the financial crisis. Financial Analysts Journal, 67(3), 51-62.
- Masood, O., Ghauri, S. M. K., & Aktan, B. (2016). Predicting Islamic banks performance through CAMELS rating model. Banks & bank systems, (11, Iss. 3), 37-43.
- Mennawi, A. N. A., & Ahmed, A. A. (2020). The Determinants of Liquidity Risk in Islamic Banks: A Case of Sudanese Banking

Sector. International Journal of Islamic Banking and Finance Research, 4(1), 38-49.

- Milošević-Avdalović, S. (2018). The impact of bank-specific factors on the liquidity of commercial banks in Serbia. *Ekonomika preduzeća*, 66(3-4), 257-265.
- Milošević-Avdalović, S., & Kalaš, B. (2016). Determinants of deposit potential as inverse liquidity indicator of commercial banks in Serbia. *Industrija*, 44(3), 61-76.
- Mohamad, A. A. S., Mohamad, M. T., & Samsudin, M. L. (2013). How Islamic banks of Malaysia managing liquidity? An emphasis on confronting economic cycles. International Journal of Business and Social Science, 4(7).
- Mounira, B. A., & Anas, E. L. M. E. L. K. I. (2008). Managing risks and liquidity in an interest free banking framework: the case of the Islamic banks. International Journal of Business and Management, 3(9), 80-95.
- Moyer, S. E. (1990). Capital adequacy ratio regulations and accounting choices in commercial banks. Journal of accounting and economics, 13(2), 123-154.
- Mwangi, M., Muturi, W., & Ombuki, C. (2015). The effects of deposit to asset ratio on the financial sustainability of deposit taking micro finance institutions in Kenya,11(2), 60-74.

- Oino, I. (2014). The nexus between bank capital, liquidity and the business cycles: Empirical evidence from the UK banking sector. International Journal of Finance and Accounting Studies, 2(1), 11-20.
- Ojo, M. (2010). Risk management by the Basel Committee: Evaluating progress made from the 1988 Basel Accord to recent developments. Journal of Financial Regulation and Compliance, 18(4), 305-315.
- Otieno, S., Nyagol, M. and Onditi, A., 2016. Empirical Analysis on Relationship between Liquidity risk management and financial performance of microfinance banks in Kenya. Research Journal of Finance and Accounting, 7(6), pp.115-142
- Palestine Monetary Authority. (2016). Instructions No. 7 of 2016. Retrievedfrom<u>https://www.pma.ps/Portals/0/Users/002/02/2/Legislati</u> on/Instructions/Banks/2016/Instructions-07-2016.pd
- Palestine Monetary Authority. (2018). Financial Stability Report 2018. Retrievedfrom<u>https://www.pma.ps/Portals/0/Users/002/02/2/Publicati</u> ons/English/Annual%20Reports/Financial%20Stability%20Reports/F <u>SR-En-2018.pd</u>
- Palestine Monetary Authority. (2019). Financial Stability Report 2019. Retrievedfrom<u>https://www.pma.ps/Portals/0/Users/002/02/2/Publicati</u> ons/English/Annual%20Reports/Financial%20Stability%20Reports/F <u>SR-En-2018.pd</u>

- Palestine Monetary Authority. (2019). PMA Achievement Report 2019. Retrievedfrom<u>https://www.pma.ps/Portals/0/Users/002/02/2/Publicati</u> ons/English/PMA%20Achievement%20Reports
- Pyle, D. H. (1999). Bank risk management: theory. In *Risk Management and regulation in banking* (pp. 7-14). Springer, Boston, MA.
- Rahman, M. L., & Banna, S. H. (2015). Liquidity risk management: a comparative study between conventional and Islamic banks in Bangladesh. *Journal of Business and Technology (Dhaka)*, 10(2), 18-35.
- Rajhi, W., Hassairi, S. A., & CERGAM, I. (2012). Capital Structure and Financial Risks in Non-Conventional Banking System. International journal of economics and finance, 4(4), 252-265.
- Rehman, M. Z., Khan, M. N., & Khokhar, I. (2015). Investigating liquidityprofitability relationship: Evidence from companies listed in Saudi stock exchange (Tadawul). *Journal of Applied Finance and Banking*, 5(3), 159.
- Rosly, S. A. (2005). Islamic banking: Doing things right and doing right things. *Malaysian Journal of Economic Studies*, 42(1), 31-40.
- Rosly, S. A., & Zaini, M. A. M. (2008). Risk-return analysis of Islamic banks' investment deposits and shareholders' fund. Managerial Finance, 34(10), 695-707..

- Rosman, R. (2009). Risk management practices and risk management processes of Islamic banks: a proposed framework. *International Review of Business Research Papers*, 5(1), 242-254.
- Said, R. M., & Tumin, M. H. (2011). Performance and financial ratios of commercial banks in Malaysia and China. International Review of Business Research Papers, 7(2), 157-169.
- Saita, F. (1999). Allocation of Risk Capital in Financial Institutions. Financial Management, 28(3), 95-95.
- Santomero, A. M., & Babbel, D. F. (1997). Financial risk management by insurers: An analysis of the process. Journal of Risk & Insurance, 64(2), 231-270.
- Shaheen & Sabah, & Misbah, B. (2011). The impact of risk management on the degree of safety in the Palestinian banking system. Al-Aqsa University Journal (Humanities Series), 15 (1), 1-2
- Soylu, Uğurlu. "Status of Islamic Finance in Germany in 2019." Islamic Finance in Europe. Palgrave Macmillan, Cham, 2019. 149-158.
- Sweeney, R. (2019). Transformation of banking reconsidered: how feasible is 'de-financialisation'?. Cambridge Journal of Economics, 43(4), 1053-1071.
- Tarawneh, M. (2006). A comparison of financial performance in the banking sector: Some evidence from Omani commercial

banks. International Research Journal of Finance and Economics, 3(3), 101-112.

- Van Greuning, H., & Iqbal, Z. (2007). Banking and the risk environment. Islamic Finance: The Regulatory Challenge, John Wiley & Sons, Singapore,11, 11-39.
- Wanjohi, S. M., Wanjohi, J. G., & Ndambiri, J. M. (2017). The effect of financial risk management on the financial performance of commercial banks in Kenya. International Journal of Finance and Banking Research, 3(5), 70-81.
- Yaacob, S. F., Rahman, A. A., & Karim, Z. A. (2016). The determinants of liquidity risk: A panel study of Islamic banks in Malaysia. *Journal of Contemporary Issues and Thought*, 6, 73-82.
- Yuwonoputro, D. A., & Syaichu, M. (2019). INDONESIAN BANKS
 RISK-TAKING: THE EFFECT OF LIQUIDITY RISK, CAPITAL
 BUFFER AND BOPO: Z-SCORE MEASURE
 ASPPROACH. Diponegoro Journal of Management, 8(3), 149-160.
- Zheng, H. (2006). Interaction of credit and liquidity risks: Modelling and valuation. Journal of Banking & Finance, 30(2), 391-407.
- Žuk-Butkuvienė, A., Vaitulevičienė, D., & Staroselskaja, J. (2014). Capital adequacy (solvency) and liquidity risk management: analysis, evaluation, and possibilities for improvement. Ekonomika, 93, 59-76.

Annexes

Annex (1)

Questionnaire to Islamic Banks

Liquidity risk Managements practices in Palestinian banks

Questionnaire number.....

This questionnaire is to assess liquidity risk management practices for Islamic Banks. The survey is designed to get information from Islamic Bankers. I would be grateful if you could take some of your time to fill out this questionnaire

1- What are the reasons behind the existence of liquidity risk in Palestinian Islamic?

2-how liquidity risk is measured by your bank?

3- If your bank has a committee to manage liquidity risk, what is the role of this committee?

Annex (2)

Questionnaire to Conventional Banks

Liquidity Risk Managements practices in Palestinian banks

Questionnaire number.....

This survey is to assess liquidity risk management practices for conventional Banks. The survey is designed to get information from conventional Bankers. I would be grateful if you could take some of your time to fill out this questionnaire

1-What are the reasons behind the existence of liquidity risk in Palestinian conventional banks?

2-how liquidity risk is measured by your bank?

3- If your bank has a committee to manage liquidity risk, what is the role of this committee?

جامعة النجاح الوطنية كلية الدر اسات العليا

ممارسات ادارة مخاطر السيوله :دراسه مقارنه بين البنوك التقليديه والاسلاميه في فلسطين



إشراف د. رأفت الجلاد

قدمت هذه الأطروحة استكمالاً لمتطلبات الحصول على درجة الماجستير في التمويل بكلية الدراسات العليا في جامعة النجاح الوطنية في نابلس، فلسطين. 2020م ممارسات ادارة مخاطر السيوله :دراسه مقارنه بين البنوك التقليديه والاسلاميه في فلسطين

تواجه البنوك أنواعًا عديدة من المخاطر منها مخاطر السيولة التي تعتبر من أهم القضايا في القطاع المصرفي (التجاري والإسلامي)، حيث يخسر البنك عددًا من عملائه نتيجة نقص السيولة الكافية أو عدم القدرة للوفاء بسحوباتهم أو أموالهم في الوقت المناسب وكذلك مساءلة السلطات الإشرافية، مما يؤثر على جانب السمعة المصرفية للمؤسسة المالية، في المقابل، قد يحتفظ البنك بسيولة أكثر مما يحتاج، مما يؤدي إلى الاستخدام غير الصحيح للموارد المتاحة، وبالتالي فإن وجود إدارة سيولة مناسبة يزيد من ثقة السلطات و المودعين.

يمكن للبنوك التقليدية استخدام العديد من االادوات لإدارة هذا النوع من المخاطر . ولكن بالمقابل تعتبر البنوك الإسلامية مقيدة بشكل متزايد في استخدام جزء من هذه الأدوات. بهذه الطريقة، من الأهمية بمكان فهم فكرة مخاطر السيولة في البنوك الإسلامية وما هي المتغير ات التي يمكن أن تؤثر وتضيف إلى مخاطر السيولة. تم استخدام منهجية البحث المختلط، حيث تـم تنفيذ المنهج النوعي من خلال عينة مكونة من 3 بنوك إسلامية و 3 بنوك تجارية ، وقـد تـم استخدام استبيان مفتوح للحصول على البيانات المستهدفة من البنوك الإسلامية والتجارية. كما تم استخدام منهجية البحث الكمي من خلال قياس مخاطر السيولة في البنوك الإسلامية والتجارية. كما تم أستخدام منهجية البحث الكمي من خلال قياس مخاطر السيولة في البنوك الإسلامية والتجاريــة في فلسطين. من خلال فحص العوامل المختلفه التي اؤثر على مخاطر السيوله: مثل نسبة كفاية رأس المال (CAR) ، والعائد على حقوق الملكية (ROE) كمؤشر لمخاطر السيولة ، نسـبة الودائع إلى الأصول (DA) وحجم البنك (BS) من البنوك الإسلامية والتجاريــة الودائع إلى الأصول (DA) وحجم البنك (BS) من البنوك الإسلامية والتجاريــة الودائع إلى الأصول (DA) وحجم البنك (BS) من البنوك الإسلامية والتجاريــة الودائع إلى المال (EAS) ، والعائد على حقوق الملكية (EOS) كمؤشر لمخاطر السيولة ، نسـبة الودائع إلى الأصول (DA) وحجم البنك (BS) من البنوك الإسلامية والتجارية. فلسطين الفترة الودائع إلى الأصول (DA) وحجم البنك (BS) من البنوك الإسلامية والتجارية. فلسطين الفترة الودائع إلى المول (DA) وحجم البنك (DA) ما البنوك الإسلامية والتجارية. فلسطين الفترة الودائع إلى الأصول (DA) وحجم البنك (BS) من البنوك الإسلامية والتجارية. فلسطين الفترة والودائع إلى المول (DA) وحجم البنك (BS) من البنوك الإسلامية والتجارية. فلسطين الفترة الودائع إلى المول (DA) وحجم البنك (BS) من البنوك الإسلامية والتجارية. فلسطين الفترة الودائع إلى الأمول (DA) وحجم البنك (BS) من البنوك الإسلامية والتجارية. فلسطين الفترة الوركة مولي والحفاط والمولية موسطين الفترة مالمي والولية ما مولي والمؤسل ما مولي والولية مالمي والولية ما مولي والولي المؤسل ما م